Author's response to reviews

Title: The silence of MUC2 mRNA induced by promoter hypermethylation associated with HBV in Hepatocellular Carcinoma

Authors:

Yang Ling (lingyangcz@126.com)
Chenghui Zhang (zhangchny@126.com)
Yun Xu (xuyunny@126.com)
Changtai Zhu (zhuuctcz@126.com)
Jing Zhu (zhuijingcz@126.com)
Yongping Liu (liuypcz@126.com)
Mingzhu Lu (lumzcz@126.com)
Changsong Zhang (hbzcs@126.com)

Version: 5 Date: 5 December 2012

Author's response to reviews:

Dear, Journal Editorial Office

We agree with the suggestion of Journal Editorial Office and strictly comply with the provisions of authorship, and respect the rights of the author.

(1) The three authors (Rong Li, Lu Gao and Lixin Wei) are still included as authors.

(2) The three new authors (Chenghui Zhang, Yun Xu, Mingzhu Lu) have not contributed to the current study and therefore cannot be considered as authors.

(3) The two authors (Changtai Zhu, Yongping Liu) should be considered as authors because they carried out a bit of experiments about this manuscript.